Author's response to reviews

Title: Primary adenocarcinoma of the stomach in Von Recklinghausen's disease (NF-1) with high serum levels of multiple tumor markers

Authors:

Kazuya Kato (pippuclinikkato@gold.ocn.ne.jp)
Atsushi Nagase (nag-ayomomi@ezweb.ne.jp)
Kazuhiko Onodera (kzon@bell.ocn.ne.jp)
Minoru Matsuda (nen@coral.ocn.ne.jp)
Yoashiaki Iwasaki (yiwasaki@cc.okayama-u.ac.jp)
Yuraino Kato (kimmy015@tea.ocn.ne.jp)
Kimitaka Kato (2shin@pippuclinic.pippu.hokkaido.jp)
Takako Kawakami (takakotk@ezweb.ne.jp)
Masahiko Taniguchi (m-tani@asahikawa-med.ac.jp)
Hiroaki Furukawa (f-hiro@asahikawa-med.ac.jp)

Version: 6 Date: 21 September 2011

Author's response to reviews: see over
Dear Dianne Pangan,

Journal of Editorial Office
BioMed Central

MS: 1846157740545768
“Primary adenocarcinoma of the stomach in Von Recklinghausen’s disease (NF-1) with high serum levels of multiple tumor markers:
a case report”

Thank you very much for your reviewing my manuscript and giving several suggestions.

We have corrected the manuscript according to the editorial team suggestions, as follows;

#1. I have blanked out the sentence of “but no lymph node enlargement was identified”, which is irrelevant information about figure 1B.

Thank you very much for your kindly arrangements.

If something wrong still exists in the revised manuscript, please let me know.

Respectfully Yours,

Department of Surgery
Pippu Clinic
Kazuya Kato, M.D., Ph.D.